Author's response to reviews

Title: Effects of a Novel Magnetic Orthopedic Appliance (MOA-III) on Dentofacial Complex in Mild Skeletal Class III Children

Authors:

Ning Zhao (zhaon1995@126.com)
Jing Feng (helen_fj@126.com)
Zheng Hu (orthohu@126.com)
Rongjing Chen (rongjingchen@126.com)
Gang Shen (gangshen@126.com)

Version: 2
Date: 21 January 2015

Author's response to reviews: see over
Dear Editor,

According to the email of Formatting required-Head & Face Medicine (1794035679157034), we made some changes to our manuscript. Below are the changes one by one:

1. We added the name of ethics committee who approved this clinical research involving human subjects and the reference number in Methods section.

2. We added a new title page with detailed information of all the authors.

3. We stated in the Methods section that all the patients and their parents or guardians have signed the written consent letters of the participation in this research.

4. We stated that the patient we presented in this paper also signed the photo release letter by his parents.

5. We have made sure the identity of the patient was fully concealed in all figures.

Thank you so much for giving us the advices to format this paper.

We would be grateful if you could consider the paper towards publication in Head & Face Medicine.

Sincerely yours,
Gang Shen